Author's response to reviews

Title: Response rates to a mailed survey of a representative sample of cancer patients randomly drawn from the Pennsylvania Cancer Registry: incentive and length effects.

Authors:

Bridget J. Kelly (bkelly@rti.org)
Taressa K. Fraze (tfraze.fraze@gmail.com)
Robert C. Hornik (rhornik@asc.upenn.edu)

Version: 6 Date: 29 June 2010

Author's response to reviews: see over
06/29/10

Melissa Norton, M.D.
BMC Medical Research Methodology
c/o BioMed Central Ltd, Floor 6
236 Gray's Inn Road
London, WC1X 8HL, United Kingdom.

Dear Dr. Norton,

Attached is a revised version of the manuscript, entitled: *Response rates to a mailed survey of representative cancer patients: incentive and length effects.*

The declaration of competing interests has been revised to reflect the editors’ suggestion.

We appreciate your consideration of these revisions.

Thank you,

Bridget Kelly, PhD, MPH
RTI Research Analyst
RTI International
701 13th Street, NW, Ste. 750
Washington, DC 20005-3967
phone: (202) 728-2098
fax: (202) 974-7855